Author’s response to reviews

Title: Frankincense oil derived from Boswellia carteri induces tumor cell specific cytotoxicity

Authors:

Mark B Frank (frank@omrf.org)
Qing Yang (qing-yang@ouhsc.edu)
Jeanette Osban (osbanj@omrf.org)
Joseph T Azzarello (joseph-azzarello@ouhsc.edu)
Marcia R Saban (marcia-saban@ouhsc.edu)
Ricardo Saban (ricardo-saban@ouhsc.edu)
Richard A Ashley (richard-ashley@ouhsc.edu)
Jan C Welter (jc-welter@ouhsc.edu)
Kar-Ming Fung (karming-fung@ouhsc.edu)
Hsueh-Kung Lin (hk-lin@ouhsc.edu)

Version: 4 Date: 9 March 2009

Author’s response to reviews:

Affiliations for all authors have been written in full.

Additional files have been included with descriptions of these files.

Keywords and Running title have been removed.

Frank et al. has been deleted from the head of every page.

e-mail address has been included for authors.

Materials and Methods has been renamed.

Page breaks have been removed in some sections.